**Author's response to reviews**

**Title:** 2 Novel Deletions of the Sterol 27-hydroxylase Gene in a Chinese Family with cerebrotendinous xanthomatosis

**Authors:**

Tian Di Jr (tiandi320@163.com)  
Zhang Zaiqiang Dr (zhangzq007@sohu.com)

**Version:** 3  **Date:** 4 September 2011

**Author's response to reviews:**

"E-mail address of Authors: DT: tiandi320@163.com" was added.

Acknowledgements

"Written consent was obtained from the patient or their relative for publication of study." was added.

5. "J Intern Med 2007 May;261(5):504-10." was corrected.